Title: Public stigma against family members of people with mental illness: Findings from the Gilgel Gibe Field Research Center (GGFRC), Southwest Ethiopia

Authors:

Eshetu Girma (grm_sht@yahoo.com)
Anne Maria Möller-Leimkühler (Anne-Maria.Moeller-Leimkuehler@med.uni-muenchen.de)
Norbert Müller (Norbert.Mueller@med.uni-muenchen.de)
Sandra Dehning (Sandra.Dehning@med.uni-muenchen.de)
Guenter Froeschl (Froeschl@lrz.uni-muenchen.de)
Markos Tesfaye (tesmarkos@yahoo.com)

Version: 2 Date: 8 November 2013

Author's response to reviews:

Correction for type errors are made